Author's response to reviews

Title: Human papillomavirus (HPV) type 16 E7 protein bodies cause tumour regression in mice

Authors:

Mark Whitehead (WHTMAR016@myuct.ac.za)
Peter Öhlschläger (oehlschlaeger@fh-aachen.de)
Fahad N Almajhdi (majhdi@KSU.EDU.SA)
Leonor Alloza (lalloza@erabiotech.com)
Pablo Marzábal (pmarzabal@erabiotech.com)
Ann E Meyers (Ann.Meyers@uct.ac.za)
Inga I Hitzeroth (Inga.Hitzeroth@uct.ac.za)
Edward P Rybicki (Ed.Rybicki@uct.ac.za)

Version: 3
Date: 10 April 2014

Author's response to reviews: see over
The Editor
BMC Cancer

Dear Editor,

Re: Revisions to Human papillomavirus (HPV) type 16 E7 protein bodies cause tumour regression in mice

The following sentences were added at the end of the document after the conclusion

Conflict of interest

This work was supported by a contract between ERA Biotech SA and the Department of Molecular and Cell Biology, University of Cape Town. Part of the results of this work is subject matter of the patent application WO2010040847.

The authors chose not to deposit the DNA into the suggested repositories as both the complete ZERA and E7SH sequences have previously been published in the following two papers:


Yours sincerely

Inga I. Hitzeroth